Author’s response to reviews

Title: Clear lens phacoemulsification in anterior lenticonus due to Alport syndrome: a case series

Authors:

Ghassem Amiraslanzadeh (dr.aslanzadeh@gmail.com)
Davood Gharabaghi (gharabaghid@gmail.com)
Niloofar Naderi (dr_niloofarnaderi@yahoo.com)

Version: 13 Date: 2 April 2008

Author’s response to reviews:

Cover letter MS: 2305466161546167
Clear lens phacoemulsification in anterior lenticonus due to Alport syndrome: a case series
I agree corrections and amendments made,
I checked and corrected references as below:
-Page 8
References
   [http://www.emedicine.com/med/topic110.htm]
   Checked
   Checked

*I would like to include these two small changes too:

Page 2
-Line 2: Introduction: Please delete Introduction it is repeated in line 21
-Line 14: Conclusion: Please add Results&:
Results& Conclusion